Author's response to reviews

Title: Transcranial sonography for diagnosis of Parkinson's disease

Authors:

Sabine Mehnert (sabine.schmidt@neuro.med.uni-giessen.de)
Iris Reuter (iris.reuter@neuro.med.uni-giessen.de)
Karsten Schepp (karsten.schepp@hotmail.de)
Peter Maaser (pemaa@gmx.de)
Erwin Stolz (erwin.stolz@neuro.med.uni-giessen.de)
Manfred Kaps (manfred.kaps@neuro.med.uni-giessen.de)

Version: 3 Date: 17 January 2010

Author's response to reviews: see over
Dear Editor-in-Chief,

we would like to thank you for acceptance of our submission, "Transcranial sonography for diagnosis of Parkinson’s disease". We have revised the manuscript according to your letter.

Figures: The figures 1-3 are now referenced in the manuscript [methods].

References: Reference 22 is now cited in the manuscript.

Header is removed.

Authors' qualifications are deleted from the manuscript and submissions system.

Keywords are removed from the title page.

Patients and methods is renamed 'Methods'.

Acknowledgements are added.

I hope that the manuscript is now ready for publication in BMC Neurology.

Sincerely,

Sabine Mehnert.